Author's response to reviews

Title: Case report of an incidental giant renal oncocytoma.

Authors:

Anastasios AA Anastasiadis (dranastasiadis@yahoo.gr)
Georgios GD Dimitriadis (gdimit@auth.gr)
Dimitrios DR Radopoulos (dradopoulos@hotmail.gr)

Version: 6 Date: 13 July 2010

Author's response to reviews:

1. The list of authors in the manuscript is exactly the same as they are in the submission system.
2. Reference [1] is removed from the abstract.
3. Figure 1 is cited in the main manuscript.
4. Ref 6 is removed from the manuscript.
5. The title has changed (as instructed).